**Author’s response to reviews**

**Title:** Sexual and reproductive health of women living with HIV in Muslim-majority countries: A systematic mixed studies review

**Authors:**
Dyah Juliastuti (d.juliastuti@uq.edu.au)
Judith Dean (j.dean4@uq.edu.au)
Lisa Fitzgerald (l.fitzgerald@sph.uq.edu.au)

**Version:** 4 **Date:** 21 Jan 2020

**Author’s response to reviews:**
Dear Editor of BMC International Health and Human Rights,

We tried our best to address all of your technical comments, including:

- revising unclear sentences and grammatical errors
- corrected inappropriate casual language and sweeping statements
- adjusted the references as suggested by the guidelines for the authors of this journal
- reviewed by someone who has high fluency in English.

We hope it meets the journal standard for publication now.

Looking forwards your response soon. Thank you.

Kind regards,

Dyah Juliastuti
Dr Judith Dean
Dr Lisa Fitzgerald